Author's response to reviews

Title: Differential expression of anterior gradient gene AGR2 in prostate cancer

Authors:

Erin L Maresh (emaresh@mednet.ucla.edu)
Veil Mah (ymah@mednet.ucla.edu)
Mohammad Alavi (malavi@mednet.ucla.edu)
Steve Horvath (shorvath@mednet.ucla.edu)
Lora Bagryanova (l_bagryanova@yahoo.com)
Emily S Liebeskind (emslieb@gmail.com)
Laura A Knutzen (laurak2@u.washington.edu)
Yong Zhou (yzhou@systemsbiology.org)
David Chia (dchia@mednet.ucla.edu)
Alvin Y Liu (aliu@u.washington.edu)
Lee Goodglick (lgoodglick@mednet.ucla.edu)

Version: 3 Date: 2 December 2010

Author's response to reviews: see over
November 29, 2010

Dr Ingunn Holen
BMC Cancer

Dear Dr. Holen:

We have made all the revisions suggested to the manuscript entitled: Differential Expression anterior gradient gene AGR2 in prostate cancer.

Specifically, we made the following changes:

1. We have added each author's full affiliation and addresses.
2. We have indicate that all authors read and approved the final manuscript.
3. We have removed shading from the tables.
4. We have removed the running title
5. We have removed blank page 14.
6. We are now consistent in the text and figure legends when citing figures.
7. The figure / images do not include titles.
8. The figures are now cropped as closely as possible to minimize white space around the image.
9. Visible vertical lines have been removed from our tables
10. We have corrected some typographical errors.

Thank you very much. Let me know if you need additional changes

Sincerely,

Lee Goodglick, Ph.D.
Associate Professor
Department of Pathology and Laboratory Medicine
David Geffen School of Medicine at UCLA